Author's response to reviews

Title: Do practice characteristics explain differences in morbidity estimates between general practice registration networks?

Authors:

Catharina van den Dungen (c.vandendungen@hotmail.nl)
Nancy Hoeymans (nancy.hoeymans@rivm.nl)
Marjan van den Akker (marjan.vandenakker@maastrichtuniversity.nl)
Marion CJ Biermans (m.biermans@radboudumc.nl)
Kees van Boven (Kees.vanBoven@radboudumc.nl)
Hanna Joosten (JHK.Joosten@vumc.nl)
Robert A Verheij (r.verheij@nivel.nl)
Margot WM de Waal (m.w.m.de_waal@lumc.nl)
Francois G Schellevis (f.schellevis@nivel.nl)
Hans van Oers (hans.van.oers@rivm.nl)

Version: 2 Date: 21 May 2014

Author's response to reviews:

Dear editors,

You asked me to revise my manuscript on two aspects;

1. the main manuscript includes line-numbers and page numbers

2. Ethics: As this research has no experimental design, this study doesn't need approval of an ethics committee. We used only anonymous secondary data of general practice registration networks (GPRN). The representatives of the GPRNs all authorized the use of the data (are also co-author of this manuscript).

If you require more information, please contact me.
Yours sincerely,

C. van den Dungen